Author’s response to reviews

Title: IgG4-related kidney disease from the renal pelvis that mimicked urothelial carcinoma: A case report

Authors:

Hui Zhang (moshanghuizi@126.com)
Xinyu Ren (renxinyu7957@163.com)
Wen Zhang (lovemexieh@126.com)
Di Yang (punchzhanghui@126.com)
Ruie Feng (fengruie1@163.com)

Version: 4 Date: 28 March 2015

Author’s response to reviews: see over
RESPONSE LETTER

Title: IgG4-related kidney disease from the renal pelvis that mimicked urothelial carcinoma: A case report

Author(s): Hui Zhang et al.

MS ID: 5402741553102223

MS Type: Case report

Email address: moshanghuizi@126.com; fengruie1@163.com

Full postal address: Department of Pathology, Peking Union Medical College Hospital, Chinese Academy of Medical Science, 1 Shuaifu Yuan, Beijing 100730, PR China

Phone: +86 010-69159356.

Dear editor,

We highly appreciate you for reviewing our paper. We have made revisions according to the comments in the revised manuscript. These changes will not influence the content and framework of the paper. Thank you so much.

Yours sincerely,

Hui Zhang
Response to editor's comments

EDITORIAL REQUEST

Comments to the Author:

1. Copyedit:
We recommend that you ask a native English speaking colleague to help you copyedit the paper. If this is not possible, you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central recommends Edanz (www.edanzediting.com/bmc1). BioMed Central has negotiated a 10% discount to the fee charged to BioMed Central authors by Edanz. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication. For more information, see our FAQ on language editing services at http://www.biomedcentral.com/info/authors/authorfaqs#12.

Response:
According to your kind suggestion, we employed Edanz to thoroughly copyedit our manuscript for language usage, spelling, and grammar.
List of Amendments